Author's response to reviews

Title: A randomized trial of sodium-restriction on kidney function, fluid volume and adipokines in CKD patients

Authors:

Katrina L Campbell (Katrina_Campbell@health.qld.gov.au)
David W Johnson (David_Johnson@health.qld.gov.au)
Judith D Bauer (j.bauer1@uq.edu.au)
Carmel M Hawley (Carmel_Hawley@health.qld.gov.au)
Nikky M Isbel (Nikky_Isbel@health.qld.gov.au)
Michael Stowasser (m.stowasser@uq.edu.au)
Goce Dimeski (Goce_Dimeski@health.qld.gov.au)
Jonathan P Whitehead (jwhitehead@mmri.mater.org.au)
Emma J McMahon (e.hall5@uq.edu.au)

Version: 3 Date: 25 February 2014

Author's response to reviews:

Thank you for the revisions to the manuscript. Your manuscript has now been assessed by our Associate Editor. Most of the issues have been well addressed. However, there are a few remaining places for improvement. Please see our editorial points (below):

Remarks:

- I would like to ask the authors to provide the exact values of central blood pressure, since these are lacking. Only the changes in central BP are given, which aren't very informative when one doesn't know baseline central BP.

Response: This has been actioned and values for systolic and diastolic central BP are now provided

- Literature reference nr. 24 is cited incompletely. The author names were not given.

Response: This has now been updated